cambridge.org/sus

Review Article

Cite this article: Kelly A (2024). A panacea to unsustainable consumption? A review of resource caps. *Global Sustainability* **7**, e18, 1–13. https://doi.org/10.1017/sus.2024.16

Received: 7 March 2023 Revised: 23 January 2024 Accepted: 18 March 2024

Keywords:

economics; industrial activities; natural resources (biological and non-biological); policies; politics and governance

Corresponding author:

Adam Kelly, Email: adkelly@student.unimelb.edu.au

A panacea to unsustainable consumption? A review of resource caps

Adam Kelly^{1,2} 💿

¹School of Geography, Earth and Atmospheric Sciences, University of Melbourne, Melbourne, Australia and ²Melbourne Climate Futures, University of Melbourne, Melbourne, Australia

Abstract

Non-technical summary. Many of the most pressing issues of today, such as climate change, habitat destruction, and conflict, are linked to our growing economies and the increasing amount of natural resources needed to maintain them. Current resource management policies focus on using resources more efficiently while maintaining economic growth. However, these policies have been insufficient and alternatives are needed. Resource caps are one such alternative which would directly limit resource consumption and extraction. This first review on the topic covers existing research on resource caps, the practical issues of implementation, and suggests a way forward for future policy and research.

Technical summary. Increasingly unsustainable rates of resource consumption and extraction have led to a growing discussion among researchers and environmental advocates on introducing caps on resource use. Research suggests that a reliance on efficiency-based approaches and a focus on decoupling are not sufficient to reduce ecosystem pressures, and instead alternatives such as resource caps may be needed. This article therefore provides the first comprehensive review of research on resource caps, linking them to major social science debates on resource scarcity, social metabolism, decoupling, and degrowth. Resource caps have been increasingly proposed in contemporary degrowth research, but this review found that resource caps are compatible with the agendas of those who endorse 'green growth' or 'ecomodernist' positions. Although resource caps are commonly proposed at a global level, it was found that enacting national or regional level caps is more viable, and that such caps should be developed through post-normal science and with democratic governance. However, current research does not show how resource caps can be implemented in practice, despite there being a detailed discussion on the political and social factors surrounding implementation. Future research will need to consider how, and even if, caps can function, and in what situations they are effective.

Social media summary. Capping consumption and extraction of natural resources is an alternative to current efficiency-based resource policies.

1. Introduction

Civilizations have been built on natural resources, with key resource requirements altering over the centuries. Although some access to resources will be needed in the future, a reduction in the associated carbon emissions and other forms of pollution are key goals of current global politics. Resource extraction and consumption are two of the most significant drivers of environmental change globally (Steinberger et al., 2010). They have been recognized under sustainable development goal 12: 'Ensure sustainable consumption and production patterns' (United Nations, 2015). Resource extraction and consumption has been linked to issues as diverse as climate change (Ivanova et al., 2015), conflict (e.g. Le Billon, 2001), and habitat destruction (Otero et al., 2020). Through these linkages they directly affect several planetary boundaries, including climate change, biogeochemical flows, and the erosion of biosphere integrity (Steffen et al., 2015).

To limit climate change, caps, limits, and quotas have been proposed and implemented to reduce carbon emissions (e.g. Liu et al., 2015; Zakeri et al., 2015). There is an extensive body of academic discussion on this topic. For example, Chakravarty et al. (2009) outline a possible framework for setting up national-level carbon caps through assessing the emissions of individuals. Quotas and permit systems have also been used to manage fisheries and freshwater (e.g. Cryer et al., 2016; Loch et al., 2018). Similarly, caps on resource consumption or extraction have been suggested as an approach to reduce resource use for minerals and for general material use. Resource caps have been increasingly listed as an option to deal with resource consumption, particularly from within the degrowth literature (Cosme et al., 2017). This article provides the first review of the interdisciplinary research on resource caps, as distinct from caps on pollutants such as carbon, in order to summarize and analyze the growing literature on this topic.

© The Author(s), 2024. Published by Cambridge University Press. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.



'Resource caps' generally refers to limits placed on how much of a resource can be used in a defined spatial area over a set time. They can take the form of upstream production caps limiting how much of a resource can be extracted, or as consumption caps limiting the quantity of a resource that can be used or imported (Alcott, 2010). Resource caps were proposed by ecological economists such as Herman Daly and others (e.g. Daly, 1974; Wetzel & Wetzel, 1995), who viewed them as a tool to help create a steady-state economy through reducing resource consumption and extraction, and allowing renewable resources to regenerate while encouraging substitutes for non-renewable resources (Daly, 2015). Although a relatively new idea, it does relate closely to rationing, which has been discussed extensively by post-war neoclassical economists. For example, Tobin (1952) viewed rationing as 'the replacement of a single-currency system with a multiple-currency system' (p. 521), and claimed that the major difference between these currencies is that the size of a ration is independent of labor inputs. However, a key difference in focus is that these early discussions of rationing focus more on the distribution of products, such as food coupons during a war, whereas a cap focuses more on distributing allowances to consume limited quantities. This is where more recent ration work is more relevant, for example the rationing of water use during drought in Australia (Grafton & Ward, 2008). Moving on from discussions of rationing, apart from the early work of steady-state economists, the idea of resource caps targeting non-renewable resources beyond the context of carbon caps was not discussed in detail again until an article by Alcott (2010) in which he argued for caps as they could directly target environmental impact, whereas many other policies aim to indirectly reduce environmental impact via addressing population, affluence or technology. (Alcott refers to and criticizes the popular 'I=PAT' formula, referring to impact, population, affluence, and technology, respectively.) The implementation of caps on pollution and carbon emissions has resurfaced the debate.

Although 'resource caps' is the most common name used, in the literature they have also been referred to as 'impact caps' (e.g. Alcott, 2010), 'depletion quotas' (e.g. Daly, 1974), or as 'diminishing resource caps' by Alexander and Gleeson (2019) to emphasize that mandated resource use should contract in size over time. Although quotas and permits are functionally similar terms to caps, in practice these have been used to refer primarily to renewable resources such as fish or water, whereas a cap has been used more in the context of pollutants or non-renewable resources. Resource caps have been envisaged in a variety of ways, ranging from abstract global caps (e.g. Freire-González, 2021), to caps on specific resources (e.g. Kallis & Martinez-Alier, 2010) or even per-person capping of total materials (e.g. Lettenmeier et al., 2014). There are different ideas of how caps should be implemented, whether they should be based on a scientific-expert assessment of depletion, renewal, and waste rates (e.g. Daly, 2015) or come about through democratic deliberation and agreement (e.g. Buch-Hansen & Koch, 2019; Kallis & Martinez-Alier, 2010; Schmelzer et al., 2022).

Over the last decade, caps have predominantly been discussed by scholars associated with the degrowth movement (see later). But there is little concrete discussion on what form these caps should take, what should be capped, and in what scenarios – if any – they may be appropriate. This review is the first to draw together and synthesize existing research on resource caps, and to highlight their contribution to policy and debate across several fields including geography, political ecology, human ecology, ecological economics, environmental economics, and resource economics. These debates are particularly relevant for academic debates on (a) social metabolism, (b) resource scarcity, (c) decoupling, and (d) degrowth.

Section 2 outlines the literature review methodology, and Section 3 discusses the current empirical research on caps. This includes research on the possible types of caps, current practice, and complementary policies. It will also outline some of the important social and political issues around setting and designing caps. Section 4 outlines the four areas of academic discussion that are closely linked to caps and existing linkages between these and resource caps. It then suggests future directions or synergies between each of these debates and research on resource caps. Section 5 provides recommendations for future research, which are centered around the questions of what political and institutional changes are needed for caps to be viable, and how would caps work in practice?

2. Methods

A traditional literature search combining a snowballing approach for past sources and searching citations for more recent sources was applied. As the cap literature is still relatively undeveloped, a systematic literature review or meta-review would not be useful as the sample size of viable articles would be too small. Additionally, a thorough systematic literature review was unviable due to the relevant key search terms such as 'cap' and 'quota' being used in many non-relevant contexts (e.g. dentistry and computing), making filtering of the results difficult.

Google Scholar and Scopus were searched for articles on caps using a variety of keywords such as 'cap', 'resource cap', 'impact cap', 'depletion quotas', 'extraction cap', and 'consumption cap'. I read through all relevant articles, of which 18 explicitly discussed resource caps, noting down main results, theoretical frameworks, and threads of discussion. Afterward, I grouped related findings and topics in order to structure my results and discussed in the context of relevant literature not explicitly discussing caps. The reference lists and citing articles were checked, with additional work found in ecological economics. Alcott (2010) was the main article on caps, and most relevant articles cited him. Some recent cap-related academic literature studies only mention caps in passing (see Section 4) and were hence excluded. The main focus was on contemporary work, not the early concepts nor Daly's steady-state economy.

3. Empirical evidence on caps

In this section, I will begin with a comprehensive overview of contemporary academic writing on resource caps. I will then discuss the types of suggested caps and their relative merits, before moving onto the potential issues that could arise through the implementation of resource caps, particularly in relation to setting the cap and issues of freedom and co-option. Table 1 is a nonexhaustive list on the 18 most relevant recent articles discussing caps. The year 2010 was chosen as a base as there were few articles mentioning resource caps beyond steady-state economists before Alcott (2010) resurfaced the discussion. As such, almost all recent work on resource caps cites Alcott (2010) and mostly uses similar justifications for caps, albeit without often mentioning his IPAT argument (see above).

Although resource caps can take several different forms, the three main options are (1) domestic tradable quotas, often called

Table 1. Resource cap-related peer-reviewed research since Alcott (2010)

Article	Description
Alcott (2010)	Makes a case for impact caps as opposed to traditional right-side IPAT solutions predominantly focused on technological efficiency
Kallis and Martinez-Alier (2010)	Outlines some of the issues facing caps, including institutional design and the dangers of eco-authoritarianism
Kallis (2011)	Argues that successful caps will inevitably lead to a decline in growth, and that cap-and-trade systems accepted by policymakers and the public are likely to be unambitious
Kallis et al. (2012)	Suggests caps as an option for future research
Sekulova et al. (2013)	A successful degrowth approach would include resource caps
Lettenmeier et al. (2014)	Outlines a suggestion for a resource cap on Finland's household consumption
Martinez-Alier et al. (2014)	Caps or taxes on raw material exports could help enable the transition to a more sustainable economy
Cosme et al. (2017)	Lists seven degrowth sources suggesting caps could be put on resource consumption and extraction
Freeman (2018)	Impact caps are a necessary part of a sustainable future
Heikkurinen and Bonnedahl (2019)	Caps are a promising sustainable alternative to dominant sustainability measures that focus on technological improvements
Zoellick and Bisht (2018)	Cites Alcott (2010) on impact caps as a strategy to directly address environmental impact rather than through indirect technological means
Henckens et al. (2019)	Suggests extraction quotas negotiated between small groups of companies on some critical resources
Otero et al. (2020)	Where caps occur, they should be at different levels of governance and include resources embedded in products and services
Stratford (2020)	Controls on rent-seeking and policies to redistribute rent could make resource caps more feasible
Freire-González (2021)	Advocates global limits on resource consumption using cap-and-trade
Keyßer and Lenzen (2021)	Gives caps and eco-taxes as an example of a degrowth policy that may lead to a shrinkage of GDP
Mastini et al. (2021)	Suggests resource caps as a possible component of a Green New Deal without growth
Schmelzer et al. (2022)	Supports democratically designed resource caps as an effective tool to reduce material consumption

cap-and-trade, (2) an upstream auction where a cap is applied at the site of extraction and permits to extract under the cap are auctioned, and (3) a tax on resource production or consumption combined with lump-sum payments to consumers (Alcott, 2010). There has been little discussion or research on what kind of resource cap would work best, although there is theoretical and empirical evidence supporting each of these in the case of carbon (Boyce, 2018). In practice, whether a carbon policy is introduced upstream or downstream, and whether it is more economically efficient at addressing emissions or maintaining international competitiveness depends more on the specific design of the carbon policy, rather than whether it is more like a tax or cap-and-trade (Goulder & Schein, 2013). The main difference is that cap-and-trade leads to more price volatility as changes in demand can lead to major changes in price, whereas the main advantage of cap-and-trade over taxes is that it guarantees emissions will not surpass a set limit, which could happen with taxes if prices are not set correctly (Goulder & Schein, 2013).

Resource caps are typically discussed with only brief references to distribution (e.g. Schmelzer et al., 2022), or assume a cap-and-trade scheme will operate (e.g. Freire-González, 2021). Little reference is made to existing research on carbon cap-and-trade, but many of the conclusions on carbon caps would be similar for resource caps. For instance, allocating resource-use certificates equally to people and commercial entities without a trading mechanism is generally seen as economically inefficient within the literature (Grafton & Ward, 2008). However, there are arguments that such systems can be 'fairer' (Baumol, 1982).

In addition to the three main options, an alternative downstream approach involves rationing by individuals (Alcott, 2010). For example, Lettenmeier et al. (2014) suggested a perperson annual material footprint of eight tons per person in Finland would be sustainable. Resource caps that involve individual rationing focusing on high-income households are a similar possibility. For example, within the climate literature, Jaccard et al. (2021) found that in order to meet climate targets strategies such as capping high-income household energy use will be necessary. However, a policy closely monitoring the consumption of each individual may be seen as draconian in more democratic states where individual liberty may be highly valued, and unless enacted with broad public support may draw comparisons to China's Social Credit System. Additionally, it seems doubtful that such a system could accurately account for the vastly differing environmental impacts of different resources, particularly as current measurements of environmental impact such as material footprint, ecological footprint, and environmental performance index are incongruent (Requena-i-Mora & Brockington, 2021). This is much more complicated than with carbon where at the very least caps and other climate policies can be set to achieve certain carbon emission targets following for example the Paris 2050 1.5°C target (e.g. Jaccard et al., 2021).

Existing cap-and-trade systems, such as for carbon, tend to have lenient caps and light regulation to avoid increasing the cost of environmentally damaging but politically and economically important industries (Kallis, 2011). As such there are few if any examples of policies that strictly regulate overall resource consumption to what could be viewed as a sustainable level (van den Bergh, 2011). No examples of caps on minerals were found in the literature, but there are many examples of caps on carbon, such as the European Union Emissions Trading Scheme (EU ETS) (Álvarez & André, 2015). Caps have also been applied to some pollutants such as sulfur in the United States (Keohane, 2009) and to fisheries, like the total allowable catch limits in New Zealand (Cryer et al., 2016). Meub et al. (2016) discussed the potential application of cap-and-trade to land consumption in Germany, with set amounts of tradable certificates given to municipalities to use for building projects, but this has not been implemented.

It is argued that caps could be complemented by spatially constrained bans on extraction in areas where high biodiversity is affected, also limiting associated transport infrastructure expansion in those areas (Otero et al., 2020). This would help further limit impact on ecologically sensitive communities. These already exist where conservationists have successfully campaigned for bans on mining in some national parks and reserves such as in Zimbabwe (BBC, 2020).

3.1 Potential issues for cap implementation

There is a tendency to assume or envisage resource caps applied globally and set at differing levels for different individual states. For example, Freire-González (2021) suggests there should be global limits implemented through cap-and-trade. Schmelzer et al. (2022) propose caps as national or global ceilings on resource extraction. Alcott (2010) also envisages caps as global, and on per country rather than per individual basis. The former is to avoid any free-riding, and the latter is to account for population changes. Another reason given in support of global caps is that more localized caps set at smaller scales could experience rebound effects through imports and trade (Santarius, 2012). For example, the effect of a cap in one place could be reduced if resourceintensive industries or investment funds are re-allocated to places not covered by the cap. However, unilaterally implementing caps is not a barrier to global progress toward a more sustainable global economy. Although not necessarily supporting resource caps, or even carbon cap-and-trade instead of carbon taxes, Edenhofer et al. (2015) argued that, in the case of carbon, the unilateral introduction of different emission pricing systems by states could allow for incremental progress toward international agreements. The same arguments can be applied to unilaterally enacted resource caps.

Although global caps are theoretically optimal, they are difficult to realize. It is difficult to envisage global caps on resource use when even global action on lowering carbon emissions is a divisive issue. States have different and competing national and economic interests surrounding resource governance that would complicate discussions and agreements. Considering that many resources have more localized environmental impact than carbon or ozone-depletion substances, it would be more difficult to appeal to any narrative of necessary international co-operation. Additionally, there is evidence in the case of carbon, particularly during early attempts to link European, Californian, and Quebec carbon markets, that when more than one national entity controls the market, it can be unstable and difficult to manage (Green, 2017).

Generally it has been assumed that resource caps, even if global, would be set or determined nationally or on similar scales (e.g. Alcott, 2010). The existence of large multinational companies and highly mobile wealthy elites does lend credence to the idea of caps on larger scales. On the other end of the spectrum, there is room to investigate caps on smaller scales, such as at a city or community level. However, this review follows the literature in focusing on caps at the level of a state. Although recognizing that caps at other scales may be viable and are worthy of investigation, they face a completely different set of problems to state-based caps, ranging from potentially lacking the legal, institutional, and practical power to set and enforce caps, to difficulties in tracking and monitoring capped resource flows across boundaries. Furthermore, current governance of resource use is predominantly at national or sub-national scales (Henckens et al., 2019), managed under national-scale property rights (Bringezu et al., 2016). This favors national-scale implementation of caps. Although there is no evidence in the literature of resource caps on minerals having been implemented at any scale, there are a variety of caps or permit schemes on national or sub-national scales that can be of instructional value in designing resource caps (e.g. Cryer et al., 2016; Keohane, 2009).

Unilateral implementation of extraction caps could result in complicated geopolitical outcomes. China's export restrictions on rare-earth metals implemented in 2010 is an example of the potential effects of such a unilateral approach (Klossek et al., 2016). Nations may be less likely to implement an extraction cap in case it interferes with free-trade arrangements or leads to a deterioration in partnerships. As such, in many cases caps on resource consumption of imported resources may be more politically and practically viable. For example, a hypothetical cap on a rare-earth resource used in a country that does not produce it would occur at the end of a supply chain and not affect as many actors, but would reduce a country's reliance on that resource, whereas a cap on the same resource at the point of extraction would affect many actors down the supply chain and have more potential for stoking geopolitical tension. In essence, depending on the resource, extraction caps are more likely to have more globalized in terms of reducing environmental impact and but also increasing geopolitical tension than with consumption caps.

Beyond the scale of the cap, an important question is who will determine the level of caps, and how they will do so (Kallis & Martinez-Alier, 2010). Steady-state economists (e.g. Daly, 2015) tend to adopt a discourse addressed at 'enlightened' knowledgeable policymakers and scientists. In contrast, others, particularly within the degrowth movement, see implementation as comprising values as well as science, questioning decision making and implementation models that involve sequestered policymakers and experts making decisions without transparency or democratic involvement (e.g. Kallis & Martinez-Alier, 2010). There are many examples of purely science-driven water and fishery policies that have been unsuccessful (Kallis & Martinez-Alier, 2010). The analysis of Buch-Hansen and Koch (2019) of a wealth and income cap strongly supports caps being democratically developed through participatory settings involving different societal groups, rather than being dictated by policymakers and experts. Fuchs et al. (2021) provide a discussion of potential participatory approaches and emphasize the need to treat people as holistic individuals rather than consumers to be influenced, and also emphasize the necessity of promoting 'citizen competence' to empower citizens in discussing potential societal futures and pathways.

There are many factors that could influence the level at which a cap is set. These range from a scientific assessment of the impact of a certain level of resource use, to an agreement drawing on what is politically acceptable, as seen in the case of the Paris Agreement (Geden, 2016). One common suggestion for resource or other caps is that they could relate to historical levels of consumption and environmental impact by country (Otero et al., 2020; Schmelzer et al., 2022). This is a common narrative in carbon discussions (particularly for countries in the Global South including China) arguing that countries with large historical emissions should shoulder a higher burden of emission reductions. However, most of this discussion assumes global discussion around the setting of caps, and is more relevant to pollutants with globalized impacts. In the case of resource caps, which primarily would be on resources such as minerals with more localized impacts, the most relevant factors could be different than for carbon. For example, in the case of existing permit systems such as Australia's Murray-Darling Basin Plan and New Zealand's fishing permit system, discussion centered around ensuring clear and tradable property rights for water and on transgressions and noncompliance for the former and on long-term environment and economic sustainability in the latter (Cryer et al., 2016; Loch et al., 2018).

If trying to reduce overall environmental impact, it is likely that several different caps on different resources or impacts would need to be set rather than an aggregate figure, with this rendering debate and decision making more complex, as has occurred for capping carbon emissions (Kallis & Martinez-Alier, 2010). Whether such caps should be at a fixed level, vary based on some criteria, or diminish over time has been discussed very little (e.g. Alcott, 2010; Alexander & Gleeson, 2019).

Another major issue that would hinder implementation of an aggregate cap on resource consumption is whether a cap could be enforced, as a cap that is not enforced would have little effect. An aggregate cap would be difficult to enforce as although economywide aggregate material flow analysis tools exist (e.g. Schaffartzik et al., 2014), it would still be almost impossible to monitor the extraction and movement of all materials. In terms of caps on specific resources then, enforceability would depend on three factors. The first is the institutional, legal, and practical powers of a state as it would need to have the power to implement and monitor a cap. The second is the type of resource being capped, it would need to be a resource that could be monitored at the point at which it is capped, for example this could be at the point of extraction, import, or export. The third is the distribution scheme, where for example certificate trading and auction might ensure easier monitoring and enforcing than in the case of directly allocating permits.

Alcott (2010) debated whether resource caps might infringe on individual freedom and whether such policies are politically acceptable. He claims that caps have no more risk of being used for authoritarian purposes than competing policy approaches such as taxation and setting penalties. Replying to Alcott's article, Kallis and Martinez-Alier (2010) suggested the process of setting and monitoring caps could result in eco-authoritarian and expertbased regimes, or draconian environmental regulation. They highlight that caps are particularly vulnerable to being used in this way due to their scientific complexity, vesting access to resources with the state, and suggest that popularly undesirable caps could be enforced by authoritarian regimes. Existing caps on carbon such as the EU ETS have been viewed as technocratic due to their heavy reliance on market mechanisms to solve environmental problems (Knox-Hayes & Hayes, 2014). Kallis and Martinez-Alier (2010) suggest that caps decided on through post-normal science (a collaborative process drawing on a combination of science and different interest groups) and democratic governance through communal institutions are possible, but that this process is potentially difficult and may result in caps quite different to those that have been theorized. This does raise the question of what a post-normal collaboratively designed cap would look like, and there is room for experimentation to see where this might lead.

We have established that the institutional changes needed for the implementation of a cap may be complicated (Kallis & Martinez-Alier, 2010). Structures to monitor, design, and enforce caps would be needed, especially to reduce the risk of co-option by corporations or for eco-authoritarian reasons. These could require large investment and would not necessarily by successful (Kallis & Martinez-Alier, 2010). Caps are only likely to succeed if there is significant legal or political support (Colby, 2000). Vested interests and strong links between politics and private interests make it difficult for meaningful policy change to occur (Kallis, 2011; Kallis et al., 2012).

Stratford (2020) highlights how rent-seeking behavior is an issue for any potential resource cap in a degrowing economy. She argues that scarcity induced through hard-environmental limits such as caps will increase prices, and that this will lead to a greater opportunity for rent capture. She views rent-seeking behavior as occurring when individuals and firms profit from controlling assets that are difficult to substitute, thus extracting rent and gaining far high-profit margins than would occur if an asset was more substitutable. For example, controlling land to lease to mining interests is an example of rent capture, whereas a new technological innovation is not as it could be copied in other companies. As such, she argues that rent-seeking prevention and redistribution measures are necessary to mitigate adverse social impacts of a cap, and would have the co-benefit of increasing popular support of such policies.

Due to the difficulties outlined above, Smith (2014) suggests that caps and similar hard environmental limits may not be possible without a radical societal restructuring and a move beyond capitalism. He argues that in some cases it would be simpler and more environmentally effective just to ban certain substances or practices outright. Others have also argued for the banning of excessive practices of the wealthy such as private jets or megaprojects such as hydro-power dams (Schmelzer et al., 2022). Again, such a perspective runs the risk of currently being politically and practically unfeasible in democratic nations. However, as the climate crisis worsens, and if other environmental issues become more acute, nations that have more dynamic democratic systems where new political parties and movements can emerge and take power may in the future be more likely to enact such policies than in authoritarian or in two-party democratic systems such as the United States and United Kingdom, as these policies would generally be against the entrenched interests of governments and elites.

4. Caps and social science debates

In this section, I will outline the background of the four social science debates listed in the Introduction and discuss how they link to current cap research. For each debate, I will then outline potential contributions research on caps could have to these debates, and the converse, how further developments in these debates can benefit future research on caps. I will begin with the discussion around the social metabolism, before moving on sequentially to resource scarcity, decoupling, and degrowth in turn.

4.1 The social metabolism

When it comes to discussions around social metabolism, the sustainability or lack thereof of global resource consumption has been debated for centuries. Arguably one of the first important works on this topic was Thomas Malthus's 1798 An Essay on the Principle of Population. The common interpretation of Malthus's argument is that population growth would eventually outpace agricultural production and available resources. However, a new reading suggests that Malthus was instead intent on discouraging revolutionary action seeking equality as a solution to problems of scarcity and in fact argued that continued productivity increases could under the right circumstances lead to continuous gains in agricultural productivity and population growth (Kallis, 2019). Recent discussion around global resource consumption and limits was made prominent by The Limits to Growth report of the Club of Rome (Meadows et al., 1972). Contemporary research within social and industrial ecology and ecological economics focuses on the 'social metabolism', which is the compilation of the continuous material and energy flows that allows society to function (Haberl et al., 2017; Mayer & Haas, 2016). This discussion has been ongoing owing to increasing rates of global resource extraction and consumption. For example, there has been a 30% increase in the global social metabolism in the first decade of the 21st century alone (Schaffartzik et al., 2014). Even under the most stringent degrowth futures the large-scale adoption of renewable energy will require large quantities of minerals (Gibon & Hertwich, 2014; Harmsen et al., 2013).

Resource extraction as a driving component of the social metabolism is a global issue, but it often plays out at local scales. Global demand for resources can lead to ecological distribution conflicts over resource access, ecosystem services, and of the location of pollution and waste outputs (Perez-Rincon et al., 2019). Ecological conflicts can lead to the erosion of livelihoods, conflict within and between communities, the displacement of communities, and even the death of local activists (Bringezu et al., 2016; Ide, 2015; Martinez-Alier et al., 2016; Tran et al., 2020). Ecological distribution conflicts are increasing in number as shown in the Environmental Justice Atlas project (Bringezu et al., 2016). These conflicts tend to be more potent in the Global South where there are less clearly defined property rights to land (Hilson, 2002). There is a further neo-colonial dimension due to many resource extraction activities being undertaken on the cultural lands of Indigenous groups, who are forced to interact with large corporations at a significant power disadvantage and with limited common ground in terms of culture and language (Perez-Rincon, 2006).

Current resource consumption levels are unsustainable according to social metabolic models. But policy outcomes from these findings are lacking. The general finding is that humanity needs to (a) increase the efficiency of resource use, or (b) reduce the size of metabolic flows (Fischer-Kowalski & Haberl, 2015). It is the next step, policy implementation, that could involve resource caps. Caps could aid in reducing the social metabolism (Martinez-Alier et al., 2014), and existing research on social metabolism acts as a rationale for reducing resource consumption and extraction (Schaffartzik et al., 2014). It has led to nationallevel indicators for monitoring resource use and consumption policies, and an international material flow database (Haberl et al., 2019). Such databases could be used to propose levels at which caps should be set based on historical levels as suggested by Otero et al. (2020). A detailed description of material flows can help encourage a transition to a more circular economy (Haberl et al., 2019). Proposals focusing on capping aggregate materials such as Lettenmeier et al.'s (2014) rationing proposal would likely have to use social metabolic assessment tools such as material flow analysis.

The social metabolic literature could benefit from a discussion of caps, as it provides one way to translate flow calculations into policy approaches. Although social metabolic tools are undoubtedly useful and perhaps necessary in gauging how to set caps, such tools can be scientifically complex and would require careful consideration on how to involve different layperson interest groups (Buch-Hansen & Koch, 2019; Kallis & Martinez-Alier, 2010). As such, the many social factors surrounding a cap in relation to social metabolism studies should be strongly emphasized. Such social factors could include, but are not limited to, inequality and wealth distribution, poverty and marginalization, transparency and accountability (e.g. Brand et al., 2021).

4.2 Resource scarcity

Resource scarcity and how to manage it has been discussed for decades. Hotelling (1931) was concerned with resource exhaustion and concluded that then-prevailing conservation methods such as mandating obsolete technology or implementing periodic temporary bans on extraction were economically inferior to taxation, and that taxation and interest rates should be set to optimize resource production. Much of the contemporary debate hinges on how likely resource scarcity is to occur in the near and long-term future and at what scale, whether markets, prices, and technological advancements are sufficient tools to manage scarcity, what forms scarcity may take, whether scarcity will have a significant impact across the planet, and how to manage it through policy intervention. In particular, there is an ongoing debate between thinkers such as Tilton (2003) who are 'modestly-optimistic' about future scarcity, and others for example Prior et al. (2012) who are concerned about increasing marginal costs and impending resource scarcity (e.g. Calvo et al., 2017; Wellmer & Scholz, 2018).

Tilton's (2003) idea is that increasing prices will drive technological innovation and mineral exploration, which will render more minerals economically viable to obtain. He claims that the issue is not resource depletion but is instead slowly increasing prices due to extraction becoming increasingly expensive. This position is commonly held by cornucopian thinkers such as Julian Simon who contest that human ingenuity combined with free markets have made resources less scarce, and will continue to do so (Aligica, 2009).

In contrast, Prior et al. (2012) argue that there are increasing marginal costs that constrain extraction but do not end it, with further extraction leading to increasing environmental and social costs (Memary et al., 2012; Prior et al., 2012; Wellmer & Scholz, 2018). They argue that although prices do correlate to some extent with scarcity (e.g. Tilton et al., 2018), they have proven to be a flawed proxy, for example in the cases of petroleum (Gordon et al., 2007) or fisheries (Akenji et al., 2016). Recent work in material science also undermines the confidence in human

ingenuity to solve scarcity issues through new technological breakthroughs. Out of 62 metals used frequently in the global economy, none have existing exemplary substitutes, and many commonly used metals, such as copper and manganese, have no good substitutes at all (Graedel et al., 2013). Both sides of the debate acknowledge that resource scarcity is an issue, but differ on their assessment of the severity of the problem and what needs to be done to manage it.

Ostrom (1990) demonstrated that resources can be collectively managed sustainably over the long term by different users, challenging the then-prevailing notion that commonly held resources would be subject to rapid exhaustion unless private property rights were implemented, the well-known 'tragedy of the commons'. However, as Ostrom herself acknowledged (Ostrom, 2010), there are different diverse resource systems and no universalized rules. In the case of global fisheries, the original tragedy of the common arguments may be valid, and there have been continuous, and sometimes controversial, suggestions to create property rights and systems similar to caps, for example on whales (Costello et al., 2012). Even without the application of the tragedy of the commons argument, resource caps could prove to be a useful tool for pre-emptively managing scarcity. Related policy concepts such as quotas have been used for managing scarce renewable resources in fisheries (e.g. Cryer et al., 2016) and for managing water resources such as aquifers and rivers (e.g. Loch et al., 2018). Cap-and-trade systems have been implemented to manage and limit the quantity of emitted pollutants such as sulfur dioxide and carbon (e.g. Álvarez & André, 2015; Keohane, 2009). Quotas have been suggested as an approach to managing critical minerals, perhaps agreed upon by groups of countries that mine them (Henckens et al., 2019). However, resource caps are yet to play a significant role in the debate over resource scarcity for nonrenewable resources such as minerals. This is surprising as Daly advocated resource caps as a way of encouraging the discovery and uptake of substitutes for scarce resources decades ago (Daly, 2015). That renewable resources have been managed under cap-like policies but not non-renewable resources could be due to their characteristics of being renewable or nonrenewable themselves, with their being an incentive to manage renewable resources so they can be harvested perpetually, whereas with non-renewable resources extraction cannot continue forever. Furthermore, non-renewable extractive activities tend to be predominantly centered on mining, which operates on a shorter temporal scale than activities such as fishing and farming, as mines are not intended to operate indefinitely.

As with social metabolism, caps could be of benefit as a management or policy option toward dealing with scarcity - for example, the enormous demand for lithium at present for battery technologies might require a cap on production, rather than free-market price inflation fueling risky and polluting mining operations in dryland lake beds and other environments (Babidge et al., 2019; Jerez et al., 2021). Caps could be investigated as a policy option that could help address concerns over resource criticality, supply risks, vulnerability to supply restrictions, and concerns over resource exhaustion and lack of substitutes. Caps could act to reduce reliance and pre-emptively manage important resources such as phosphorus (e.g. Alewell et al., 2020) before any issues arise. A resource cap would in essence artificially induce scarcity earlier, but in a controlled manner through which it would be easier to deal with variability in resource supply as opposed to a situation of uncontrolled scarcity.

There have been long-term predictions of extractable resource exhaustion in economic, social, and environmental terms without proper management for resources such as gold, copper, and nickel (Henckens et al., 2019). There are also concerns over the longterm exhaustion and uneven build-up of important minerals for the biosphere, such as phosphorus (e.g. Cordell et al., 2009; Ragnarsdottir et al., 2011). Market pricing is unlikely to address longer term supply concerns (Henckens et al., 2019) and capping either production or consumption of phosphate fertilizers or depletion rates would help deal with issues of disruption sooner than they will arise. Similarly, caps could apply to resources with few effective substitutes such as magnesium, manganese, and chromium (Graedel et al., 2015), to drive scientific endeavors to find possible substitutes or to drive more efficient use.

The case of China's restriction of rare-earth metals in 2010 (Klossek et al., 2016) demonstrates that geopolitical concerns should be accounted for in any discussion of resource caps in the context of resource scarcity. Although China's restriction on rare-earth metal exports was undertaken for political reasons rather than to manage scarcity on the Chinese side, it does illustrate the kinds of economic and political ramifications a cap on a scarce resource can have, as Japan responded by drawing on resource reserves and attempting to diversify supply chains. In general, a resource cap encapsulating exports could affect importing countries and therefore damage relationships. In light of geopolitical concerns, there is an important discussion to be had surrounding whether implementing caps at the point of extraction, or at the point of import or consumption, would be better approaches for managing resource scarcity. A cap that came into effect further up the supply chain or at the point of extraction would be more likely to have wider geopolitical effects. In contrast the unilateral decision to restrict imports for environmental reasons may have less geopolitical ramifications as particularly for many scarce and high-impact resources such as phosphorus and rare-earth metals there are fewer locations where such resources are extracted and exported from compared to places they are imported to. However, this is not to say that an import cap would be without geopolitical implications as seen in the case of China's recent import ban on many Australian products.

4.3 Decoupling

How necessary it is to reduce resource consumption and extraction and how best to do so is a complex and topical issue, and there are two major camps. Mainstream 'ecological modernists' such as Ted Nordhaus tend to argue that societies can decouple economic growth from resource consumption, by scientific and engineering led technological advancement (e.g. Asafu-Adjaye et al., 2015). The possibility of decoupling is central to the idea of green growth (contested by Hickel & Kallis, 2019) supported by major institutions such as the World Bank, United Nations Environment Program, and the Organization for Economic Co-operation and Development.

In an investigation of 116 countries, Hubacek et al. (2021) found that 14 countries had managed to absolutely decouple both consumption- and production-linked emissions from GDP growth. Although there has been some success with emissions, there has been less success with material consumption in general. Despite large efficiency gains in the use of materials, there has not been a corresponding total reduction and absolute decoupling of material use (Shao et al., 2017). Although absolute decoupling could occur in the future (Meyer et al., 2012), in recent years

even trends toward relative decoupling have often been reversing (Krausmann et al., 2018). For proponents of decoupling, if it is possible then it should be possible to achieve economic growth under a resource cap. As such, introducing resource caps could also be used to validate the realizability of decoupling. However, there is no empirical evidence supporting the possibility that global decoupling of resources or emissions can occur globally and at a rate necessary to limit climate change in line with the Paris goals (Hickel & Kallis, 2019). This is a major reason why policies such as caps have been suggested as options to manage consumption and emissions.

Critics of the decoupling idea, often endorsing degrowth precepts, believe that it is not possible to decouple economic growth from resource consumption at a rate sufficient to avoid loss and damage of ecosystems. This academic argument has major practical implications, as if it is not possible to decouple material consumption from economic growth then new approaches, such as those suggested by degrowth proponents, are required. Even beyond degrowth circles, related ideas have become more mainstream, with the concept of 'sufficiency' appearing in the IPCC 6th Assessment Report, which defines it as a 'set of measures and daily practices that avoid demand for energy, materials, land, and water while delivering human well-being for all within planetary boundaries' (IPCC, 2023, p. 72). If decoupling is not possible, then continuous expansion of material affluence would not be possible, and sufficiency would need to be the focus, with resource caps being a policy approach that could help achieve it.

Indeed, and especially if the decoupling argument is unconvincing, there is a need to investigate and implement more socially acceptable approaches to reducing resource consumption and extraction and managing scarcity (Gorg et al., 2019). If decoupling is not possible, then policies compatible with degrowth such as resource caps may be needed to reduce material consumption. However, even if there is a consensus supporting the introduction of policies such as resource caps, political economy constraints are important. The repeal of Australia's ETS, in essence a carbon cap, provides an illustrative case study of such issues. After being discussed since 1997, in 2007 the introduction of an ETS became the bipartisan policy of Australia's two major parties. It took a further 5 years and one failed attempt for carbon pricing to be eventually introduced in 2012, but without bipartisan support (Crowley, 2017). However, a conservative government was elected in 2013 after running a campaign targeting the ETS and repealed the program. As such, this case study shows that popular and political support can change, and that policies can take a significant time to move from conceptualization, to policy, to implementation. Furthermore, there are issues of path dependency, where even if new policies are introduced it can take time to change direction (Djelic & Quack, 2007). There has been extensive discussion on how these types of political economy issues relate to decoupling and degrowth in the literature (e.g. Djelic & Quack, 2007). As such, although alternatives to decoupling may be required, these too face significant constraints around implementation.

Caps have been discussed in the context of decoupling, specifically as a way of mitigating the rebound effect. The rebound effect is when increases in resource-use efficiency are reinvested in further resource-consuming activity and as such the originally efficiency savings are reduced (Alcott, 2005). Caps are theoretic-ally the most effective approach for reducing rebound effects as money saved from becoming more efficient at using a resource

or else cannot be reinvested into continued extraction or use of that resource (Santarius, 2012). Caps are viewed by Vivanco et al. (2016) as being more attractive than taxes as they address impact directly, rather than increasing efficiency, which does not always lead to absolute reductions in impact. Freire-González (2021) explicitly suggests cap-and-trade as a way of addressing the rebound effects of reducing resource use. However, Santarius (2012) cautions that caps that are not on a global scale may not prevent rebound through imports and trade.

The effects of implementing resource caps could have direct significance to theoretical debates surrounding decoupling. Although one of the justifications for implementing caps is the view that a sustainable rate of decoupling is impossible, the success of resource cap policies is independent of whether decoupling is possible. In other words, the success of a cap would be judged by whether material consumption does decrease regardless of whether it hinders or aids economic growth.

In fact, resource caps could provide evidence on decoupling, by acting as a litmus test. If stringent caps, whether on aggregate material use or targeting multiple resources with a highenvironmental impact were implemented in several places, the effects on GDP would form empirical data for further evaluating the decoupling hypothesis.

4.4 Degrowth and ecological modernization

Finally, whether and how we should degrow our economies is an ongoing debate. Degrowth advocates typically support resource caps and most resource cap discussion has occurred within this context. In contrast, there has been little discussion of resource caps from more mainstream ecological modernists, who have nonetheless addressed carbon caps.

Kallis (2011) defines sustainable degrowth as 'a socially sustainable and equitable reduction (and eventual stabilisation) of society's throughput' (p. 874). The degrowth movement dates from the 1970s and it is now present in activist movements and academic and policy circles (Weiss & Cattaneo, 2017). Kallis (2011) suggests that degrowth serves as a unifying keyword for a variety of different policies and initiatives that have roots in criticisms of development and the ability to decouple economic growth from resource consumption and environmental impact. Degrowth generally centers around (a) contraction of economic and material outputs in the Global North, which would allow space for (b) livelihood improvements in the Global South, entailing greater affluence and material intensity, albeit through alternative pathways to the dominant neoliberal sustainable development pathways (Hanaček et al., 2020). At least in regard to energy, Oswald et al. (2021) found that redistribution and increased equality can ensure billions of people no longer face energy poverty, and at the same time while others would need to reduce energy consumption, no one else would fall below this level.

Van den Bergh (2011) argues the degrowth literature seems to be more idealistic and utopian rather than practical. He suggests drastic societal and economic changes proposed may not in fact lead to the expected outcomes and may rather contribute to large-scale unrest and upheaval. Instead, it may be better to add new institutions and to apply concrete environmental limits, such as caps on production, to existing economic activity. Martinez-Alier et al. (2010) posit that one of the main issues with degrowth, in comparison to more mainstream movements such as sustainable development, is that it is more challenging to current political actors and that few in government or in private industries feel they could support it. Schmelzer et al. (2022) argue that planning for degrowth is a major research gap to address and that the management of resource caps are something that needs to be addressed within this context. Degrowth supporters favor caps of different kinds, because of the lack of success of the approaches mentioned above, and their support of more radical, society-changing actions (Jackson, 2009).

Most contemporary cap research has occurred within the literature on degrowth. Alexander & Gleeson (2019) suggest that diminishing resource caps are needed to degrow the material base of an economy, and may in fact lead to more efficient resource use. Cosme et al. (2017) list seven degrowth sources in a table suggesting caps, both tradable and non-tradable, on resource consumption and extraction. Typically, resource caps are presented as one of many items on a degrowth agenda, or as a component of a successful degrowth strategy (e.g. Sekulova et al., 2013). Mastini et al. (2021) also view environmental caps and bans as a degrowth strategy. Schmelzer et al. (2022) support resource caps as a more effective alternative to current market and technology-based approaches. Zoellick and Bisht (2018) cite Alcott (2010) on impact caps, and highlight them as an example of a 'sufficiency' strategy to reduce resource use often advocated for by degrowth supporters. Keyßer and Lenzen (2021) present caps and eco-taxes as an example of a degrowth policy that may lead to a shrinkage of GDP. Kallis et al. (2012) highlight that caps and other related policy options are a topic on which the degrowth and steady-state economy literature overlaps, and suggest that caps are a promising direction. Kallis and Martinez-Alier (2010) support the idea of caps as a degrowth strategy but outline many difficulties surrounding them (see Section 3). Finally, although caps have been supported in many of the articles outlined above, there is some dissonance; while most papers assume a cap-and-trade approach, many researchers who write on degrowth themes are skeptical regarding the expansion of market-based approaches (e.g. Fotopoulos, 2007, Trainer, 2014). Although resource caps using a cap-and-trade approach may entail the creation of new markets, this would be on resources that are already traded on markets. This differentiates resource caps from more problematic proposals to create markets on for instance biodiversity and ecosystems.

In contrast to degrowth, resource caps are, as noted above, absent from the ecological modernization literature. By contrast, caps on carbon and other pricing instruments are viewed as the most economically efficient tools at reducing greenhouse gas emissions (e.g. Edenhofer et al., 2015; Schmalensee & Stavins, 2017). Reasons given to support carbon caps are that they stimulate technological development and lead to a more efficient allocation of resources (Edenhofer et al., 2015). There is no research on why carbon caps are relatively mainstream among ecomodernists and resource caps are not. There are several plausible reasons. The first could be due to confidence in technology to find substitutes for scarce resources, meaning that resource scarcity could be perceived as far less immediate a threat than climate change. Second, there could simply be a lack of knowledge on the issues surrounding resource consumption compared with climate change. Third, climate change is a highly global issue with clear links between carbon emissions and climate change, whereas the impacts of resource consumption can be more localized and the links between consumption, extraction, and impact can be complex and hidden.

Resource caps are policies that bear some similarities with the EU ETS, which is a multi-nation carbon market utilizing cap-and-trade and covering many industries within the EU, and could be used as a stepping-stone to a degrowth policy agenda, or as a way of finding common ground with those who advocate alternative futures. A resource cap is arguably compatible with degrowth, zero growth, and growth approaches. However, one of the key issues with sequential and gradual policy change is that caps may be overly lenient and fail to influence strong environmental outcomes. For example, the EU ETS has been criticized for not leading to a large enough reduction in emissions, leading to unequal distribution of costs, and insufficiently driving innovation (Branger et al., 2015).

5. Recommendations for future research

Two main future research directions on resource caps emerge. The first relates to the theory behind resource caps and current socio-political environments. What kinds of political or institutional changes are needed for resource caps, and how will they be set and by whom? The second is how will caps work in practice? What kinds of resources, if any, should be capped, how, and where?

As resource caps would require extensive programs in terms of setting, monitoring, and design (e.g. Kallis & Martinez-Alier, 2010), it is unclear whether current legal and policy institutions would be suitable for implementing resource caps, although existing carbon caps such as the EU ETS could be used as an initial point of reference.

As outlined cogently by Kallis and Martinez-Alier (2010), some of the most important questions surrounding resource caps pertain to how they would be set and by whom. As with any other government policy, the power relations of access and control will determine the effectiveness of resource caps not only in mitigating environmental problems but also balancing social and economic equity (e.g. Bryant, 1998; Ribot & Peluso, 2003). As many government policies are vulnerable to co-option to some degree by powerful actors, or to (mis)use by authoritarian regimes, the question is whether resource caps are any more likely than other policies to be enlisted in this way. As seen in Section 4, it is possible resource caps may be more likely than some other policies to fall afoul of these issues. For example, large corporations could stand to make large profits by controlling resources restricted by caps (Stratford, 2020).

Although Kallis and Martinez-Alier (2010) state that most researchers take democracy as a given and are strongly against supporting authoritarian regimes in any form, there is room for a realpolitik argument, as has occurred with calls for stronger action on climate change, in favor of resource caps. There are nations such as China that have high levels of environmental impact and show no signs of transitioning to become more democratic or allow more citizen participation in governance in the near future. Researchers may need to engage with such states and proceed with research on sustainability policies regardless of the risk of co-option. Without strong cooperation, including with less democratic regimes, it will be difficult to design and implement constraints on global resource use.

There are clear issues with such a realpolitik approach. The first is the history of environmental policies being enacted by powerful elite groups and being used to justify control or displacement of minority groups, particularly Indigenous communities (e.g. Benjaminsen & Bryceson, 2012; Scheidel & Work, 2018).

As caps would limit mining, this seems less likely to occur than with other environmental policies, but it could be used as an excuse to justify government crackdown on local peoples in precarious mining situations, such as artisanal cobalt miners in the Congo. Additionally, it is questionable whether many authoritarian regimes would have a legitimate desire to pursue environmentally friendly policies, even if economically viable, if they come at economic or political cost. As authoritarian regimes rely on concessions or rent-sharing with powerful actors to maintain power (Gandhi & Przeworski, 2006), environmental policies that reduce the ability of the state to extract rent from resources would be against the political interest of most authoritarian regimes.

The second main recommendation is to investigate how caps could work in practice. As outlined above, there has been no research investigating what kinds of resources should be capped and where they could be capped, and even less experimentation with resource caps outside of those on carbon and renewable resources. Research has been at a very abstract level. Caps could target specific resources such as molybdenum, of concern to Henckens et al. (2019), or be aimed at some form of aggregate material flows, as suggested by Lettenmeier et al. (2014) in their work on rationing. The former approach may be more promising, as it would be prohibitively difficult to track, monitor, and regulate overall economy-wide material use given current monitoring abilities. In contrast, it has been shown with various degrees of success that we can cap or apply permit systems to specific substances such as water and carbon emissions.

However, deciding what resources could be worth capping is difficult. First, the work on resource criticality by Thomas Graedel and others is a promising starting point for selecting a resource as it covers environmental impact and risks to resource supply that would cover both economic and security concerns (e.g. Graedel et al., 2015). But there could be other considerations such as the cultural importance or use of resources. Second, capping specific resources, while simpler than an aggregate-material cap, is still highly complicated, as resources such as indium, neodymium, and gold can be embedded at a low value in a myriad of products and flows are complicated and can criss-cross borders (e.g. Thiébaud et al., 2018).

As a policy approach that would require extensive planning, monitoring, and regulation, resource caps would not be effective for all states or places. There are still questions surrounding what kinds of governmental, political, and social conditions would render a cap feasible, and whether there are existing institutional and legal frameworks in place, or whether such frameworks are possible, that would allow caps to occur on scales beyond the state or supranational. Caps would have to function as intended, and would need popular or legislative support to be maintained, to avoid the fate of the Australian carbon cap-and-trade scheme that was removed upon a change of government in 2013. The presence of international treaties similar to the UNFCCC may help maintain such policies. It would also need to make economic sense due to the cost of maintaining a monitoring program, and would need to lead to justifiably significant environmental or social outcomes. Kallis and Martinez-Alier (2010) suggest that there has been a move away from regulatory caps in other domains such as water policy toward more flexible, local policies, and there would need to be clear reasons for undergoing the cost and time of implementing caps at a given scale, and monitoring implementation, rather than attempting a more flexible regulatory approach.

Further research is needed into the design characteristics of resource caps. For example, what kind of redistribution or permit schemes would work best? The literature on carbon caps would provide insight here. At this stage research on resource caps is too theoretical and undeveloped to transfer easily to practical implementation. Hence the purpose of this review to summarize and outline a path forward for future research in this area, as the economic and political constraints to caps have emerged as significant.

6. Conclusion

Increasingly unsustainable rates of resource extraction and consumption are driving innovative scientific and policy responses to a growing global crisis, anchored by the need to address climatic change, carbon emissions, and biodiversity loss. Resource caps are a promising approach to reducing consumption of specific resources, including critical minerals, that have a highenvironmental impact and could be included as an option in discussions about addressing overreach in the social metabolism and resource scarcity. As the literature suggests, decoupling economic growth from adverse environmental impacts and material consumption appears impossible in practice with current levels of technology and energy sources (Trainer, 2007).

Therefore, resource caps and other 'strong' approaches to sustainable consumption and production should be considered. Caps on resources have been called for by supporters of degrowth, who aim at wasteful, highly carbon intense, and unequal economies. Pursuing further research on caps can help bridge the gap in planning for degrowth. This review has found, however, that resource caps are not incompatible with the agendas of those that support or are neutral regarding the 'weaker' sustainability implied by continued economic growth and more conscientious use of the earth's resources combined with a modernist belief in the power of technological advances.

However, there are several issues surrounding resource caps that need to be addressed. Resource caps could easily lend themselves toward technocratic approaches, as seen in the EU ETS, despite researchers calling for caps to be developed through postnormal science. Although this review suggested that caps could be implemented through the adoption of techniques from the social metabolic literature, this could further risk technocracy. While resource caps are most likely to be implemented at a national or regional level, doing so without broader international collaboration could risk geopolitical ramifications, particularly if caps are implemented further up the supply chain such as at the point of extraction. Furthermore, it is still unclear what kinds of institutional or government structures are needed to maintain and implement caps, following the urgent and still partial regimes directing global carbon emissions legislated by recent UNFCCC agreements. It is also unclear whether resource caps run a larger risk of authoritarian co-option than other policies, with potential appropriation by corporate interests or authoritarian states. More generally, the key questions of how, and even if, resource caps can work in practice remain unanswered.

Acknowledgments. I thank my supervisors, Simon Batterbury and Tobias Ide, for their invaluable and continuous support throughout my preparation and writing of this manuscript. I also thank both reviewers for their extensive and thoughtful feedback.

Author contributions. The work is solely the contribution of the primary author.

Funding statement. This research is supported by an Australian Government Research Training Program (RTP) Scholarship.

Competing interests. None.

References

- Akenji, L., Bengtsson, M., Bleischwitz, R., Tukker, A., & Schandl, H. (2016). Ossified materialism: Introduction to the special volume on absolute reductions in materials throughput and emissions. *Journal of Cleaner Production*, *132*, 1–12. https://doi.org/10.1016/j.jclepro.2016.03.071
- Alcott, B. (2005). Jevons' paradox. Ecological Economics, 54, 9–21. https://doi. org/10.1016/j.ecolecon.2005.03.020
- Alcott, B. (2010). Impact caps: Why population, affluence and technology strategies should be abandoned. *Journal of Cleaner Production*, 18, 552– 560. https://doi.org/10.1016/j.jclepro.2009.08.001
- Alewell, C., Ringeval, B., Ballabio, C., Robinson, D. A., Panagos, P., & Borrelli, P. (2020). Global phosphorus shortage will be aggravated by soil erosion. *Nature Communications*, 11, 1–12. https://doi.org/10.1038/s41467-020-18326-7
- Alexander, S., & Gleeson, B. (2019). Degrowth in the suburbs: A radical urban imaginary. Palgrave Macmillan, Springer Nature.
- Aligica, P. D. (2009). Julian Simon and the 'limits to growth' Neo-Malthusians. The Electronic Journal of Sustainable Development, 1(3), 73–84. https://www. semanticscholar.org/paper/Julian-Simon-and-the-%22Limits-to-Growth%22-Aligica-Drago%C8%99/f9865c12329e57fd65924a006c8093014c0cd391
- Álvarez, F., & André, F. J. (2015). Auctioning versus grandfathering in cap-and-trade systems with market power and incomplete information. *Environmental Resource Economics*, 62, 873–906. https://doi.org/10.1007/ s10640-014-9839-z
- Asafu-Adjaye, J., Blomquist, L., Brand, S., Brook, B. W., Defries, R., Ellis, E., Foreman, C., Keith, D., Lewis, M., Lynas, M., Nordhaus, T., Pielke, R., Pritzker, R., Roy, J., Sagoff, M., Shellenberger, M., Stone, R., & Teague, P. (2015). An ecomodernist manifesto. Retrieved from http://www. ecomodernism.org/manifesto-english
- Babidge, S., Kalazich, F., Prieto, M., & Yager, K. (2019). 'That's the problem with that lake; it changes sides': Mapping extraction and ecological exhaustion in the Atacama. *Journal of Political Ecology*, 26, 739–760. https://doi. org/10.2458/v26i1.23169
- BBC. (2020). Zimbabwe bans coal mining in Hwange and other game parks. Retrieved from https://www.bbc.com/news/world-africa-54085549
- Baumol, W. J. (1982). Applied fairness theory and rationing policy. The American Economic Review, 72(4), 639–651. https://www.jstor.org/stable/ 1810007
- Benjaminsen, T. A., & Bryceson, I. (2012). Conservation, green/blue grabbing and accumulation by dispossession in Tanzania. *The Journal of Peasant Studies*, 39(2), 335–355. https://doi.org/10.1080/03066150.2012.667405
- Boyce, J. K. (2018). Carbon pricing: Effectiveness and equity. *Ecological Economics*, 150, 52–61. https://doi.org/10.1016/j.ecolecon.2018.03.030
- Brand, U., Muraca, B., Pineault, E., Sahakian, M., Schaffartzik, A., Novy, A., Streissler, C., Haberl, H., Asara, V., Dietz, K., Lang, M., Kothari, A., Smith, T., Spash, C., Brad, A., Pichler, M., Plank, C., Velegrakis, G. Jahn, T., ... Görg, C. (2021). From planetary to societal boundaries: An argument for collectively defined self-limitation. Sustainability: Science, Practice and Policy, 17(1), 264–291. https://doi.org/10.1080/15487733.2021.1940754
- Branger, F., Lecuyer, O., & Quirion, P. (2015). The European Union emissions trading scheme: Should we throw the flagship out with the bathwater? *Wiley Interdisciplinary Reviews: Climate Change*, 6(1), 9–16. https://doi.org/10. 1002/wcc.326
- Bringezu, S., Potocnik, J., Schandl, H., Lu, Y., Ramaswami, A., Swilling, M., & Suh, S. (2016). Multi-scale governance of sustainable natural resource use – Challenges and opportunities for monitoring and institutional development at the national and global level. *Sustainability*, *8*, 1–25. https://doi.org/10. 3390/su8080778
- Bryant, R. L. (1998). Power, knowledge and political ecology in the third world: A review. Progress in Physical Geography, 22(1), 79–94. https://doi. org/10.1177/030913339802200104

- Buch-Hansen, H., & Koch, M. (2019). Degrowth through income and wealth caps? *Ecological Economics*, 160, 264–271. https://doi.org/10.1016/j. ecolecon.2019.03.023
- Calvo, G., Valero, A., & Valero, A. (2017). Assessing maximum production peak and resource availability of non-fuel mineral resources: Analyzing the influence of extractable global resources. *Resources, Conservation & Recycling*, 125, 208–217. https://doi.org/10.1016/j.resconrec.2017.06.009
- Chakravarty, S., Chikkatur, A., de Coninck, H., Pacala, S., Socolow, R., & Tavoni, M. (2009). Sharing global CO₂ emission reductions among one billion high emitters. *PNAS*, 106(29), 11884–11888. https://doi.org/10.1073/ pnas.0905232106
- Colby, B. G. (2000). Cap-and-trade policy challenges: A tale of three markets. Land Economics, 76(4), 638–658. https://doi.org/10.2307/3146957
- Cordell, D., Drangert, J. O., & White, S. (2009). The story of phosphorus: Global food security and food for thought. *Global Environmental Change*, 19, 292–305. https://doi.org/10.1016/j.gloenvcha.2008.10.009
- Cosme, I., Santos, R., & O'Neill, D. (2017). Assessing the degrowth discourse: A review and analysis of academic degrowth policy proposals. *Ecological Economics*, 149, 321–334. https://doi.org/10.1016/j.jclepro.2017.02.016
- Costello, C., Gerber, L. R., & Gaines, S. (2012). A market approach to saving the whales. *Nature*, 481, 139–140. https://doi.org/10.1038/481139a
- Crowley, K. (2017). Up and down with climate politics 2013–2016: The repeal of carbon pricing in Australia. WIREs Climate Change, 8, e458. https://doi. org/10.1002/wcc.458
- Cryer, M., Mace, P. M., & Sullivan, K. J. (2016). New Zealand's ecosystem approach to fisheries management. *Fisheries Oceanography*, 25, 57–70. https://doi.org/10.1111/fog.12088
- Daly, H. (1974). Steady-state economics versus growthmania: A critique of orthodox conceptions of growth, wants, scarcity and efficiency. *Policy Sciences*, 5, 149–167. https://www.jstor.org/stable/4603736
- Daly, H. (2015). Economics for a full world. Great Transition Institute. https:// www.inist.org/library/2015-09-04.Daly.Economics%20for%20a%20Full% 20World.GTI.pdf
- Djelic, M. L., & Quack, S. (2007). Overcoming path dependency: Path generation in open systems. *Theory and Society*, 36(2), 161–186. https://doi.org/ 10.1007/s11186-007-9026-0
- Edenhofer, O., Jakob, M., Creutzig, F., Flachsland, C., Fuss, S., Kowarsch, M., Lessman, K., Mattauch, L., Siegmeier, J., & Steckel, J. C. (2015). Closing the emission price gap. *Global Environmental Change*, *31*, 132–143. https://doi. org/10.1016/j.gloenvcha.2015.01.003
- Fischer-Kowalski, M., & Haberl, H. (2015). Social metabolism: A metric for biophysical growth and degrowth. In J. Martinez-Alier & R. Muradian (Eds.), *Handbook of ecological economics* (pp. 100–138). Edward Elgar.
- Fotopoulos, T. (2007). Is degrowth compatible with a market economy? *The International Journal of Inclusive Democracy*, 3(1), 1–16. https://www. inclusivedemocracy.org/journal/vol3/vol3_no1_Takis_degrowth.htm
- Freeman, R. (2018). A theory on the future of the rebound effect in a resourceconstrained world. Frontiers in Energy Research, 6(81), 1–13. https://doi.org/ 10.3389/fenrg.2018.00081
- Freire-González, J. (2021). Governing Jevon's paradox: Policies and systemic alternatives to avoid the rebound effect. *Energy Research & Social Science*, 71, 101893. https://doi.org/10.1016/j.erss.2020.101893
- Fuchs, D., Sahakian, M., Gumbert, T., Di Giulio, A., Maniates, M., Lorek, S., & Graf, A. (2021). Consumption corridors: Living a good life within sustainable limits. Routledge.
- Gandhi, J., & Przeworski, A. (2006). Cooperation, cooptation, and rebellion under dictatorships. *Economics & Politics*, 18(1), 1–25. https://doi.org/10. 1111/j.1468-0343.2006.00160.x
- Geden, O. (2016). The Paris Agreement and the inherent inconsistency of climate policymaking. WIREs Climate Change, 7, 790–797. https://doi.org/10. 1002/wcc.427
- Gibon, T., & Hertwich, E. (2014). A global environmental assessment of electricity generation technologies with low greenhouse gas emissions. *Procedia CIRP*, 15, 3–7. https://doi.org/10.1016/j.procir.2014.06.070
- Gordon, R. B., Bertram, M., & Graedel, T. E. (2007). On the sustainability of metal supplies: A response to Tilton and Lagos. *Resources Policy*, 32, 24–28. https://doi.org/10.1016/j.resourpol.2007.04.002

- Gorg, C., Plank, C., Wiedenhofer, D., Mayer, A., Pichler, M., Schaffartzik, A., & Krausmann, F. (2019). Scrutinizing the great acceleration: The Anthropocene and its analytic challenges for social-ecological transformations. *The Anthropocene Review*, 7, 42–61. https://doi.org/10.1177/ 2053019619895034
- Goulder, L. H., & Schein, A. R. (2013). Carbon taxes versus cap and trade: A critical review. *Climate Change Economics*, 4(3), 1–28. https://www.jstor. org/stable/climchanecon.4.3.02
- Graedel, T. E., Harper, E. M., Nassar, N. T., Nuss, P., & Reck, B. K. (2015). Criticality of metals and metalloids. *PNAS*, 112(14), 4257–4262. https:// doi.org/10.1073/pnas.1500415112
- Graedel, T. E., Harper, E. M., Nassar, N. T., & Reck, B. K. (2013). On the material basis of modern society. *PNAS*, 112(20), 6295–6300. https://doi. org/10.1073/pnas.1312752110
- Grafton, R. Q., & Ward, M. B. (2008). Prices versus rationing: Marshallian surplus and mandatory water restrictions. *The Economic Record*, 84, 57–65. https://doi.org/10.1111/j.1475-4932.2008.00483.x
- Green, J. F. (2017). Don't link carbon markets. *Nature*, 543, 484–486. https:// doi.org/10.1038/543484a
- Haberl, H., Wiedenhofer, D., Erb, K. H., Gorg, C., & Krausmann, F. (2017). The material stock-flow-service nexus: A new approach for tackling the decoupling conundrum. *Sustainability*, *9*, 1049. https://doi.org/10.3390/ su9071049
- Haberl, H., Wiedenhofer, D., Pauliuk, S., Krausmann, F., Muller, D. B., & Fischer-Kowalski, M. (2019). Contributions of sociometabolic research to sustainability science. *Nature Sustainability*, 2, 173–184. https://doi.org/10. 1038/s41893-019-0225-2
- Hanaček, K., Roy, B., Avila, S., & Kallis, G. (2020). Ecological economics and degrowth: Proposing a future research agenda from the margins. *Ecological Economics*, 169, 106495. https://doi.org/10.1016/j.ecolecon.2019.106495
- Harmsen, J. H. M., Roes, A. L., & Patel, M. K. (2013). The impact of copper scarcity on the efficiency of 2050 global renewable energy scenarios. *Energy*, 50, 62–73. https://doi.org/10.1016/j.energy.2012.12.006
- Heikkurinen, P., & Bonnedahl, K. J. (2019). Dead ends and living futures: A framework for sustainable change. In K. J. Bonnedahl & P. Heikkurinen (Eds.), Strongly sustainable societies: Organising human activities on a Hot and full earth (pp. 289–301). Routledge.
- Henckens, M. C. L. M., Biermann, F. H. B., & Driessen, P. P. J. (2019). Mineral resources governance: A call for the establishment of an international competence center on mineral resources management. *Resources, Conservation* & *Recycling, 141, 255–263.* https://doi.org/10.1016/j.resconrec.2018.10.033
- Hickel, J., & Kallis, G. (2019). Is green growth possible? *New Political Economy*, 25(4), 469–486. https://doi.org/10.1080/13563467.2019.1598964
- Hilson, G. (2002). An overview of land use conflicts in mining communities. Land Use Policy, 19, 65–73. https://doi.org/10.1016/S0264-8377(01)00043-6
- Hotelling, H. (1931). The economics of exhaustible resources. Journal of Political Economy, 39(2), 137–175. https://www.jstor.org/stable/1822328
- Hubacek, K., Chen, X., Feng, K., Wiedmann, T., & Shan, Y. (2021). Evidence of decoupling consumption-based CO₂ emissions from economic growth. *Advances in Applied Energy*, 4, 100074. https://doi.org/10.1016/j.adapen. 2021.100074
- Ide, T. (2015). Why do conflicts over scarce renewable resources turn violent? A qualitative comparative analysis. *Global Environmental Change*, *33*, 61–70. https://doi.org/10.1016/j.gloenvcha.2015.04.008
- Ivanova, D., Stadler, K., Steen-Olsen, K., Wood, R., Vita, G., Tukker, A., & Hertwich, E. G. (2015). Environmental impact assessment of household consumption. *Journal of Industrial Ecology*, 20(3), 526–536. https://doi. org/10.1111/jiec.12371
- IPCC. (2023). Synthesis Report of the IPCC Sixth Assessment Report (AR6): Longer Report. https://report.ipcc.ch/ar6syr/pdf/IPCC_AR6_SYR_LongerReport.pdf
- Jaccard, I. S., Pichler, P. P., Többen, J., & Weisz, H. (2021). The energy and carbon inequality corridor for a 1.5°C compatible and just Europe. *Environmental Research Letters*, 16, 064082. https://doi.org/10.1088/1748-9326/abfb2f
- Jackson, T. (2009). Prosperity without growth: Economics for a finite planet. Earthscan.
- Jerez, B., Garcés, I., & Torres, R. (2021). Lithium extractivism and water injustices in the Salar de Atacama, Chile: The colonial shadow of green

electromobility. Political Geography, 87, 102382. https://doi.org/10.1016/j. polgeo.2021.102382

- Kallis, G. (2011). In defence of degrowth. *Ecological Economics*, 70, 873–880. https://doi.org/10.1016/j.ecolecon.2010.12.007
- Kallis, G. (2019). Why Malthus was wrong and why environmentalists should care. Stanford University Press.
- Kallis, G., & Martinez-Alier, J. (2010). Caps yes, but how? A response to Alcott. Journal of Cleaner Production, 18, 1570–1573. https://doi.org/10. 1016/j.jclepro.2010.06.010
- Kallis, G., Kerschner, C., & Martinez-Alier, J. (2012). The economics of degrowth. *Ecological Economics*, 84, 172–180. https://doi.org/10.1016/j. ecolecon.2012.08.017
- Keohane, N. O. (2009). Cap and trade, rehabilitated: Using tradeable permits to control U.S. greenhouse gases. *Review of Environmental Economics and Policy*, 3(1), 42–62. https://doi.org/10.1093/reep/ren021
- Keyßer, L. T., & Lenzen, M. (2021). 1.5°C Degrowth scenarios suggest the need for new mitigation pathways. *Nature Communications*, 12, 2676. https://doi. org/10.1038/s41467-021-22884-9
- Klossek, P., Kullik, J., & van den Boogaart, K. G. (2016). A systemic approach to the problems of the rare earth market. *Resources Policy*, 50, 131–140. https://doi.org/10.1016/j.resourpol.2016.09.005
- Knox-Hayes, J., & Hayes, J. (2014). Technocratic norms, political culture and climate change governance. *Geografska Annaler: Series B, Human Geography*, 96(3), 261–276. https://www.jstor.org/stable/43299500
- Krausmann, F., Lauk, C., Haas, W., & Wiedenhofer, D. (2018). From resource extraction to outflows of wastes and emissions: The socioeconomic metabolism of the global economy, 1900–2015. *Global Environmental Change*, 52, 131–140. https://doi.org/10.1016/j.gloenvcha.2018.07.003
- Le Billon, P. (2001). The political ecology of war: Natural resources and armed conflicts. *Political Geography*, 20(5), 561–584. https://doi.org/10.1016/ S0962-6298(01)00015-4
- Lettenmeier, M., Liedtke, C., & Rohn, H. (2014). Eight tons of material footprint – Suggestion for a resource cap for household consumption in Finland. *Resources*, 3, 488–515. https://doi.org/10.3390/resources3030488
- Liu, L., Chen, C., Zhao, Y., & Zhao, E. (2015). China's carbon-emissions trading: Overview, challenges and future. *Renewable and Sustainable Energy Reviews*, 49, 254–266. https://doi.org/10.1016/j.rser.2015.04.076
- Loch, A., Wheeler, S. A., & Settre, C. (2018). Private transaction costs of water trade in the Murray-Darling Basin. *Ecological Economics*, 146, 560–573. https://doi.org/10.1016/j.ecolecon.2017.12.004
- Martinez-Alier, J., Anguelovski, I., Bond, P., Del Bene, D., Demaria, F., Gerber, J. F., Greyl, L., Haas, W., Healy, H., Marin-Burgos, V., Ojo, G., Porto, M., Rijnhout, L., Rodriguez-L, B., Spangenberg, J., Temper, L., Warlenius, R., & Yánez, I. (2014). Between activism and science: Grassroots concepts for sustainability coined by environmental justice organizations. *Journal of Political Ecology*, 21, 20–60. https://doi.org/10.2458/v21i1.21124
- Martinez-Alier, J., Demaria, F., Temper, L., & Walter, M. (2016). Changing social metabolism and environmental conflicts in India and South America. *Journal of Political Ecology*, 23, 468–491. https://doi.org/10.2458/ v23i1.20252
- Martinez-Alier, J., Pascual, U., Vivien, F. D., & Zaccai, E. (2010). Sustainable de-growth: Mapping the context, criticisms and future prospects of an emergent paradigm. *Ecological Economics*, 2010, 1741–1747. https://doi. org/10.1016/j.ecolecon.2010.04.017
- Mastini, R., Kallis, G., & Hickel, J. (2021). A green new deal without growth? *Ecological Economics*, 179, 106382. https://doi.org/10.1016/j.ecolecon.2020. 106832
- Mayer, A., & Haas, W. (2016). Cumulative material flows provide indicators to quantify the ecological debt. *Journal of Political Ecology*, 23, 350–363. https://doi.org/10.2458/v23i1.20222
- Meadows, D. H., Meadows, D. L., Randers, J., & Behrens III, W. W. (1972). The limits to growth; a report for the club of Rome's project on the predicament of mankind. Universe Books.
- Memary, R., Giurco, D., Mudd, G., & Mason, L. (2012). Life cycle assessment: A time-series analysis of copper. *Journal of Cleaner Production*, 33, 97–108. https://doi.org/10.1016/j.jclepro.2012.04.025
- Meub, M., Proeger, T., Bizer, K., & Henger, R. (2016). Experimental evidence on the resilience of a cap & trade system for land consumption in Germany.

Land Use Policy, 51, 95-108. https://doi.org/10.1016/j.landusepol.2015.10. 018

- Meyer, B., Meyer, M., & Distelkamp, M. (2012). Modeling green growth and resource efficiency: New results. *Mineral Economics*, 24, 145–154. https:// doi.org/10.1007/s13563-011-0008-3
- Ostrom, E. (1990). Governing the commons: The evolution of institutions for collective action. Cambridge University Press.
- Ostrom, E. (2010). Beyond markets and states: Polycentric governance of complex economic systems. *American Economic Review*, 100, 641–672. https:// www.jstor.org/stable/27871226
- Oswald, Y., Steinberger, J. K., Ivanova, D., & Millward-Hopkins, J. M. (2021). Global redistribution of income and household energy footprints: A computational thought experiment. *Global Sustainability*, *4*, 1–14. https://doi. org/10.1017/sus.2021.1
- Otero, I., Farrell, K. N., Pueoyo, S., Kallis, G., Kehoe, L., Haberl, H., Plutzar, C., Hobson, P., García-Márquez, J., Rodríguez-Labajos, B., Martin, J. L., Erb, K. H., Schindler, S., Nielsen, J., Skorin, T., Settele, J., Essl, F., Gómez-Baggethun, E., Brotons, L., ... Pe'er, G. (2020). Biodiversity policy beyond economic growth. *Conservation Letters*, 13(4), 1–18. https://doi.org/10.1111/conl.12713
- Pérez-Rincón, M. A. (2006). Colombian International trade from a physical perspective: Towards an ecological 'Prebisch thesis'. *Ecological Economics*, 59, 519–529. https://doi.org/10.1016/j.ecolecon.2005.11.013
- Pérez-Rincón, M. A., Vargas-Morales, J., & Martinez-Alier, J. (2019). Mapping and analysing ecological distribution conflicts in Andean countries. *Ecological Economics*, 157, 80–91. https://doi.org/10.1016/j.ecolecon.2018.11.004
- Prior, T., Giurco, D., Mudd, G., Mason, L., & Behrisch, J. (2012). Resource depletion, peak minerals and the implications for sustainable resource management. *Global Environmental Change*, 22, 577–587. https://doi.org/10. 1016/j.gloenvcha.2011.08.009
- Ragnarsdottir, K. V., Sverdrup, H. U., & Koca, D. (2011). Challenging the planetary boundaries I: Basic principles of an integrated model for phosphorus supply dynamics and global population size. *Applied Geochemistry*, 26, 303–306. https://doi.org/10.1016/j.apgeochem.2011.03.088
- Requena-i-Mora, M., & Brockington, D. (2021). Seeing environmental injustices: The mechanics, devices and assumptions of environmental sustainability indices and indicators. *Journal of Political Ecology*, 28(1), 1023– 1050. https://doi.org/10.2458/jpe.4765
- Ribot, J. C., & Peluso, N. L. (2003). A theory of access. *Rural Sociology*, 68(2), 153–181. https://doi.org/10.1111/j.1549-0831.2003.tb00133.x
- Santarius, T. (2012). Green growth unravelled: How rebound effects baffle sustainability targets when the economy keeps growing. Green Growth Knowledge Platform. https://www.greengrowthknowledge.org/research/ green-growth-unravelled-how-rebound-effects-baffle-sustainability-targetswhen-economy
- Schaffartzik, A., Mayer, A., Gingrich, S., Eisenmenger, N., Loy, C., & Krausmann, F. (2014). The global metabolic transition: Regional patterns and trends of global material flows, 1950–2010. *Global Environmental Change*, 26, 87–97. https://doi.org/10.1016/j.gloenvcha. 2014.03.013
- Scheidel, A., & Work, C. (2018). Forest plantations and climate change discourses: New powers of 'green' grabbing in Cambodia. *Land Use Policy*, 77, 9–18. https://doi.org/10.1016/j.landusepol.2018.04.057
- Schmalensee, R., & Stavins, R. N. (2017). The design of environmental markets: What have we learned from experience with cap and trade? Oxford Review of Economic Policy, 33(4), 572–588. https://doi.org/10.1093/oxrep/ grx040
- Schmelzer, M., Vetter, A., & Vansintjan, A. (2022). The future is degrowth: A guide to a world beyond capitalism. Verso.
- Sekulova, F., Kallis, G., Rodríguez-Labajos, B., & Schneider, F. (2013). Degrowth: From theory to practice. *Journal of Cleaner Production*, 38, 1– 6. https://doi.org/10.1016/j.jclepro.2012.06.022
- Shao, Q., Schaffartzik, A., Mayer, A., & Krausmann, F. (2017). The high 'price' of dematerialization: A dynamic panel data analysis of material use and

economic recession. Journal of Cleaner Production, 167, 120-132. https://doi.org/10.1016/j.jclepro.2017.08.158

- Smith, R. (2014). Beyond growth or beyond capitalism? Truthout. https:// truthout.org/articles/beyond-growth-or-beyond-capitalism/
- Steffen, W., Richardson, K., Rockström, J., Cornell, S. E., Fetzer, I., Bennett, E. M., Biggs, R., Carpenter, S. R., De Vries, W., De Wit, C. A., Folke, C., Gerten, D., Heinke, J., Mace, G. M., Persson, L. M., Ramanathan, V., Reyers, B., & Sörlin, S. (2015). Planetary boundaries: Guiding human development on a changing planet. *Science (New York, N.Y.)*, 347(6223), 736– 748. https://doi.org/10.1126/science.1259855
- Steinberger, J. K., Krausmann, F., & Eisenmenger, N. (2010). Global patterns of material use: A socioeconomic and geophysical analysis. *Ecological Economics*, 69, 1148–1158. https://doi.org/10.1016/j.ecolecon.2009.12.009
- Stratford, B. (2020). The threat of rent extraction in a resource-constrained future. *Ecological Economics*, 169, 1–11. https://doi.org/10.1016/j.ecolecon. 2019.106524
- Thiébaud, E., Hilty, L. M., Schluep, M., Böni, H. W., & Faulstich, M. (2018). Where do our resources go? Indium, neodymium, and gold flows connected to the use of electronic equipment in Switzerland. *Sustainability*, 10, 1–17. https://doi.org/10.3390/su10082658
- Tilton, J. E. (2003). On borrowed time? Assessing the threat of mineral depletion. Minerals & Energy – Raw Materials Report, 18(1), 33–42. https://doi. org/10.1080/14041040310008383
- Tilton, J. E., Crowson, P. C. F., DeYoung Jr, J. H., Eggert, R. G., Ericsson, M., Guzman, J. I., Humphreys, D., Lagos, G., Maxwell, P., Radetzki, M., Singer, D. A., & Wellmer, F. W. (2018). Public policy and future mineral supplies. *Resources Policy*, 57, 55–60. https://doi.org/10.1016/j.resourpol.2018.01.006
- Tobin, J. (1952). A survey of the theory of rationing. *Econometrica*, 20(4), 521– 553. https://doi.org/10.2307/1907642
- Trainer, T. (2007). Renewable energy cannot sustain a consumer society. Dordrecht: Springer.
- Trainer, T. (2014). The degrowth movement from the perspective of the simpler way. *Capitalism, Nature, Society, 26*(2), 58–75. https://doi.org/10.1080/ 10455752.2014.987150
- Tran, D., Martinez-Alier, J., Navas, G., & Mingorria, S. (2020). Gendered geographies of violence: A multiple case study analysis of murdered women environmental defenders. *Journal of Political Ecology*, 27(1), 1189–1212. https://doi.org/10.2458/v27i1.23760
- United Nations. (2015). Transforming our world: The 2030 Agenda for Sustainable Development. https://www.un.org/ga/search/view_doc.asp? symbol=A/RES/70/1&Lang=E
- van den Bergh, J. C. J. M. (2011). Environment versus growth A criticism of 'degrowth' and a plea for 'a-growth'. *Ecological Economics*, 70, 881–890. https://doi.org/10.1016/j.ecolecon.2010.09.035
- Vivanco, D. F., Kemp, R., & van der Voet, E. (2016). How to deal with the rebound effect? A policy-oriented approach. *Energy Policy*, 94, 114–125. https://doi.org/10.1016/j.enpol.2016.03.054
- Weiss, M., & Cattaneo, C. (2017). Degrowth Taking stock and reviewing an emerging academic paradigm. *Ecological Economics*, 137, 220–230. https:// doi.org/10.1016/j.ecolecon.2017.01.014
- Wellmer, F. W., & Scholz, R. W. (2018). Peak gold? Not yet! A response to Calvo et al. (2017). Resources, Conservation and Recycling, 134, 313–314. https://doi.org/10.1016/j.resconrec.2017.11.015
- Wetzel, K. R., & Wetzel, J. F. (1995). Sizing the earth: Recognition of economic carrying capacity. *Ecological Economics*, 12, 13–21. https://doi.org/10.1016/ 0921-8009(94)00019-R
- Zakeri, A., Dehghanian, F., Fahimnia, B., & Sarkis, J. (2015). Carbon pricing versus emissions trading: A supply chain planning perspective. *International Journal of Production Economics*, 164, 197–205. https://doi.org/10.1016/j.ijpe. 2014.11.012
- Zoellick, J. C., & Bisht, A. (2018). It's not (all) about efficiency: Powering and organizing technology from a degrowth perspective. *Journal of Cleaner Production*, 197, 1787–1799. https://doi.org/10.1016/j.jclepro.2017.03.234